Author's response to reviews

Title: The clinical effectiveness of an ultra-brief intervention for common mental health syndromes in primary care: study protocol for a cluster randomised controlled trial

Authors:

Sunny C Collings Prof (sunny.collings@otago.ac.nz)
Fiona Mathieson Snr Lecturer (fiona.mathieson@otago.ac.nz)
Anthony Dowell Prof (tony.dowell@otago.ac.nz)
James Stanley Dr (james.stanley@otago.ac.nz)
Simon Hatcher Dr (shatcher@uottawa.ca)
Felicity A Goodyear-Smith Prof (f.goodyear-smith@auckland.ac.nz)
Brigitte Lane Dr (stress.study@otago.ac.nz)
Amy Munsterman (amyemunsterman@gmail.com)

Version: 7 Date: 21 May 2015

Author's response to reviews:

Third revision changes:
1. Date study registered and trial registration number included at the end of the Abstract
2. Each author individually mentioned in the Authors’ Contributions section, including SH and AM who were not previously included in this section.
3. Additional files with responses to reviewer comments removed.